Author's response to reviews

Title: A European perspective of Salmonella enterica subps. arizonae: a case report.

Authors:

Stefano Di Bella (stefano932@gmail.com)
Alessandro Capone (alessandro.capone@inmi.it)
Eugenio Bordi (eugenio.bordi@inmi.it)
Emma Johnson (emmas@hotmail.com)
Maria Musso (maria_musso@hotmail.it)
Simone Topino (simone.topino@inmi.it)
Pasquale Noto (pasquale.noto@inmi.it)
Nicola Petrosillo (nicola.petrosillo@inmi.it)

Version: 5 Date: 15 April 2011

Author's response to reviews: see over
The JMCR Editorial Team

April 15, 2011

MS: 4314542955008953
A European perspective of *Salmonella enterica* subsp. *arizonae*: a case report.
Stefano Di Bella *et al.*

Dear Editor:

Attached please find the text of our manuscript, revised according to the Journal style.

Below you will find a reply to Editorial Support Assistant.

Do not hesitate to contact me for further details

Kind Regards

Stefano Di Bella

Stefano Di Bella, M.D.
National Institute for Infectious Diseases
“Lazzaro Spallanzani” – Via Portuense, 292-Rome, Italy
Tel +39 0655170432  fax +39 0655170486
e-mail stefano.dibella@inmi.it
Reply to Editorial Support Assistant:

- Inform us whether Table 2 is original to your case report, and if not please provide a reference.

  *Reply: we confirm that Table 2 is original to our case report*

- In the case presentation, please change the date ‘2008’ to the period of time before the case presentation that the patient relapsed

  *Reply: we changed it in the text*